Author's response to reviews

Title: Lack of Association Between Chlamydia Pneumoniae Serology and Endothelial Dysfunction of Coronary Arteries

Authors:
Markus Ferrari (markus.ferrari@med.uni-jena.de)
Gerald S Werner (gerald.werner@med.uni-jena.de)
Barbara M Richartz (info@kliche-richartz.de)
Albrecht Oehme (albrecht.oehme@med.uni-jena.de)
Eberhard Straube (eberhard.straube@med.uni-jena.de)
Hans R Figulla (hans.figulla@med.uni-jena.de)

Version: 4 Date: 26 April 2005

Author's response to reviews:

MS: 2847005566293952
Lack of Association Between Chlamydia Pneumoniae Serology and Endothelial Dysfunction of Coronary Arteries
Markus Ferrari, Gerald S Werner, Barbara M Richartz, Albrecht Oehme, Eberhard Straube and Hans R Figulla
Cardiovascular Ultrasound

Dear Professor Picano,
the format of our manuscript was changed according to the formatting changes requested by the journal production staff. We are proud that you accepted our manuscript for publication in Cardiovascular Ultrasound.

Sincerely Yours,
Markus Ferrari
Ass. Professor of Internal Medicine

PD Dr. Dr. Markus Ferrari
Oberarzt Klinik fur Innere Medizin I
Friedrich - Schiller - Universitat
Erlanger Allee 101
D - 07740 Jena
Tel. +49 (0)3641 - 9324527
FAX +49 (0)3641 - 9324102
Email: markus.ferrari@med.uni-jena.de